Author's response to reviews

Title: Prevalence of metabolic syndrome in Murcia Region, a southern European Mediterranean area with low cardiovascular risk and high obesity

Authors:

Diana Gavrila (diana.gavrila@carm.es)
Diego Salmerón (diego.salmeron@carm.es)
José-Manuel Egea-Caparrós (josem.egea3@carm.es)
José M Huerta (jmhuerata.carm@gmail.com)
Alfonso Pérez-Martínez (alfonso.perez3@carm.es)
Carmen Navarro (carmen.navarro@carm.es)
María-José Tormo (mjose.tormo@carm.es)

Version: 5 Date: 8 July 2011

Author's response to reviews:

I have introduced minor changes to the manuscript that concern only the format of the tables.